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@ NuMI-MINOS Status Report — T. vang

* FAR DETECTOR: Alec Habig
* One DAQ hard drive swapped out on Wed. Nov. 21
* 99.8% livetime this week

* NEAR DETECTOR: Alec Habig

* Downtime due to work on the cooling plumbing at the
downstream end, serving both the absorber and the ND.
Tue. Nov. 20

* The ND operated without coil Tuesday night as the LCW
water was de-ionized. The coil was back up at ~9:00am on
Wednesday morning.

* ND electronics hardware swaps Tue. Nov. 20 and Wed. Nov.
21 - most of the ND electronics problems were fixed.

* 97.3% livetime this week
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 BEAM SYSTEM STATUS: Sam Childress

2+5 mixed mode operation - 2.2 sec. cycle time giving 145 - 150
kW (pre shutdown typically 175 - 180 kW)

Improved beam power to 185kW this afternoon!
3.28x10*® POT for this week

Downtimes: _‘ y f
~ 36 hr downtime (Mon. 4 AM - Tues 4 PM) & "  This nut worked

"\ loose due to arcing.
i

due to oil leak in kicker power supply -
thru high voltage feed-thru.

In parallel problem with absorber RAW skid
cooling water flow - fixed

Steady operation thru holiday weekend with minimal downtime - thanks
operation staff!


http://www-bd.fnal.gov/cgi-mcr/elog.pl?nb=2007&action=view&page=-4612&button=yes&invert=yes
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Beam power plot shows booster tuning
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Intensity plot shows booster tuning
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Total NuMI protons to 00:00 Monday 26 November 2007
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Neutrino energy spectra at ND J. Hartnell
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* Plot made a few hours ago

« Have to check calibration, b-field, etc.
 The difference between two spectra is partly due to the fact that the
decay pipe is now filled with 1atm helium gas.
« Full detector simulation is on the way.
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One beam neutrino candidate at FD, found by Ryan Nichol on Wed. Nov. 22

Date : 22 Nov 2007 Time :02:14:27 Run:39987 6 Snarl : 79730 EventType : Golden Beam Neutrino

trigger : SPILL IP




